Author's response to reviews

Title: A Retrospective Analysis of Health Systems in Denmark and Kaiser Permanente

Authors:

Anne AF Frølich (anne.frolich@dadlnet.dk)
Michaela MLS Schiøtz (m.schiottz@pubhealth.ku.dk)
Martin MSL Strandberg-Larsen (m.strandberg-larsen@pubhealth.ku.dk)
John JH Hsu (john.hsu@kp.org)
Allan AK Krasnik (krasnik@pubhealth.ku.dk)
Finn FD Diderichsen (f.diderichsen@soomed.ku.dk)
Jim JB Bellows (jim.bellows@kp.org)
Jes JS Søgaard (jes@dsi.dk)
Karen KW White (kwhite@psg.ucsf.edu)

Version: 4 Date: 6 November 2008

Author's response to reviews:

I changed the title by removing the following text ‘Learning from a Comparison of Apples and Oranges’ as proposed by the Assistant Editor.
I hereby clarify that the data from the Danish Health Care System is openly available. We do not need ethical approval in DK for using this kind of register data.